Tutorial #1 (for week 3) (Jan 21, 2003)

15th January 2003

The following questions will be considered during the tutorial sessions on Tuesday...

Q1: Find an example of a CGI delivered application/form on the Internet. Give the URL, a brief description, and state why you believe it is a CGI-based application.

Q2: Find an example of a GUI blooper, or hall-of-shame entry. Briefly describe both it, and the fixed version. (Before and after pictures/screenshots and brief description).

Q3: Research: What is Norman’s^1 “Theory of Action”, and what possible practical use can it have? Q4: For the X code on page 36, for each call to the X toolkit (Xt) give a brief description of the function of the call, along with it’s parameters.

If there is time, the tutor will spend some time helping with Assignment #1.

1No relation!